

RESPONSE TO REVIEWERS

July 2, 2013



Dear Editor,

I would like to thank you for reviewing our manuscript. The reviewers' comments about our manuscript have been considered and suggested revisions were made.

Title: Lymphedema and Rheumatologic Disorders **ID(00505835)**

Author: Sibel Eyigor, MD

Name of Journal: World Journal of Rheumatology

ESPS Manuscript NO: 3676

The manuscript was improved according to the suggestions of reviewers:

1. Format was updated.

Abstract has been rewritten upon the suggestion of the editor and reviewer. Core tip has been added. The language of the paper has been revised by the author and a competent native speaker of English.

2. Revision has been made according to the suggestions of the reviewer.

Reviewer Comments:

00505788

- The figure about lympho-scintigraphy was not added because consent could not be obtained from the patient. Explanatory information about the treatment was added. In addition, Baker cyst in diagnosis was added to the diagnosis part. Language check was performed.

00460875

- Additional information about the treatment was included.

00458932

* Abstract was rewritten upon the suggestion of the editor and reviewer. The manuscript was reviewed throughout. The second half of the introduction was moved to pathogenesis section. Answers to the reviewer's questions were included in the manuscript. Treatment and conclusion sections were rearranged.

3. References and typesetting were corrected

Thank you again for publishing our manuscript in the World Journal of Rheumatology

Sincerely yours,

A handwritten signature in black ink, appearing to be 'Sibel Eyigor', written over a horizontal line.

Sibel Eyigor, MD

Associate Prof.

Ege University Medical Faculty Physical Medicine and Rehabilitation Department

35100 Bornova/Izmir/Turkey

Tel: +902323903687 Fax: +902323881953, e-mail: eyigor@hotmail.com